Author’s response to reviews

Title: Understanding the socio-economic and sexual behavioural correlates of male circumcision across eleven voluntary medical male circumcision priority countries in southeastern Africa

Authors:

- Fiona K. Lau (fionakylau@hotmail.com)
- Sylvia Jayakumar (sylvia.jayakumar@gmail.com)
- Sema K. Sgaier (Sema.Sgaier@gatesfoundation.org)

Version: 2
Date: 16 December 2014

Author’s response to reviews:

Dear Krisha,

In regard to the request on minor changes to the manuscript, please find revisions below:

Line 65: "No human subject research was involved."

Acknowledgements section (p. 18): "This study was supported with funding from the Bill and Melinda Gates Foundation." Please also note that no other contributors provided substantial support to this study/manuscript; hence, no mention was needed.

Please find the revised manuscript with the above changes attached.

Many thanks,

Fiona